Author’s response to reviews

Title: Preventing stress-related ill health among newly registered nurses by supporting engagement in proactive behaviors: Development and feasibility testing of a behavior change intervention

Authors:

Elin Frögéli (elin.frogeli@ki.se)
Ann Rudman (ann.rudman@ki.se)
Brjánn Ljótsson (brjann.ljotsson@ki.se)
Petter Gustavsson (petter.gustavsson@ki.se)

Version: 3 Date: 04 Dec 2017

Author’s response to reviews:

Dear Mr Mbuagbaw

Thank you for reviewing our paper and giving us the opportunity to improve it to fit the standards of publications in Pilot and Feasibility Studies. We have found the comments from the reviewers very helpful. We would especially like to thank the reviewers for the level of specificity in their stated concerns. This greatly facilitated the process of revising the manuscript.

We hope that you will find our revisions satisfactory. We have included all comments from the reviewers together with our responses in a table that we have attached as a supplement (“Rebuttal”). In the manuscript, we have marked all changes in yellow.

Sincerely,

The authors through Elin Frögéli